



Article Who Should Not Be Surveilled for HCC Development after Successful Therapy with DAAS in Advanced Chronic Hepatitis C? Results of a Long-Term Prospective Study

Alessia Ciancio [®], Davide Giuseppe Ribaldone [®], Matteo Spertino, Alessandra Risso, Debora Ferrarotti, Gian Paolo Caviglia [®], Patrizia Carucci, Silvia Gaia, Emanuela Rolle, Marco Sacco [®] and Giorgio Maria Saracco *[®]

> Gastro-Hepatoloy Unit, Department of Medical Sciences, University of Turin, 10126 Turin, Italy * Correspondence: giorgiomaria.saracco@unito.it; Tel.: +39-011-6336397

Abstract: Background and aims: The identification of patients with Hepatitis C Virus (HCV)-positive advanced chronic liver disease (aCLD) successfully treated by Direct Acting Antiviral Agents (DAAs) who really benefit from Hepatocellular Carcinoma (HCC) surveillance programs is still a matter of debate. We performed a long-term prospective cohort study on F3-F4 HCV-positive patients achieving Sustained Virologic Response (SVR) after DAAs treatment in order to identify patients who can safely suspend surveillance. Methods: 1000 patients with HCV-positive aCLD obtaining SVR by DAAs from January 2015 to December 2017 were divided into four groups according to baseline elastographic, ultrasonographic, clinical and biochemical features: (1) Group 1: 324 patients with Liver Stiffness Measurement (LSM) $\ge 9.5 \le 14.5$ kPa, FIB-4 < 3.25 and APRI < 1.5 (2) Group 2: 133 patients with LSM \ge 9.5 \le 14.5 kPa, FIB-4 \ge 3.25 and/or APRI \ge 1.5 (3) Group 3: 158 patients with LSM > 14.5 kPa, FIB-4 < 3.25 and APRI < 1.5 (4) Group 4: 385 patients with LSM > 14.5 kPa, FIB-4 \geq 3.25 and/or APRI \geq 1.5. FIB-4 and APRI scores were calculated at baseline and at SVR achievement. Each patient was surveiled twice-yearly by ultrasound for a median follow-up of 48 months. Results: among Group 1 patients, 1/324 (0.3%) developed HCC (0.09/100 patients/year [PY]), compared to 6/133 (4.5%) Group 2 patients (1.22/100 PY, *p* = 0.0009), 10/158 (6.3%) Group 3 patients (1.68/100 PY, *p* = 0.0001), 54/385 (14.0%) Group 4 patients (4.01/100 PY, *p* < 0.0001). HCC incidence was significantly lower in Group 2 compared to Group 3 (p = 0.004) and in Group 3 compared to Group 4 (p = 0.009). HCC risk fell in patients showing a decrease of FIB-4/APRI scores. Conclusions: the risk of HCC occurrence is negligible in about 90% of HCV-positive patients with baseline LSM \ge 9.5 \le 14.5 kPa plus FIB-4 < 3.25 and APRI < 1.5 achieving SVR. Among this particular subset of patients, FIB-4/APRI scores may represent an accurate and inexpensive tool to distinguish patients not needing long-term HCC surveillance.

Keywords: Hepatocellular Carcinoma; chronic hepatitis C; DAA; cirrhosis; FIB-4

1. Lay Summary

International guidelines are divergent regarding the long-term HCC surveillance in F3 patients cured from HCV infection. The European Association for the Study of the Liver (EASL) recommends twice yearly abdominal ultrasounds in both F3 and F4 patients, while the American Association for the Study of the Liver Diseases (AASLD) recommends surveillance only in those with cirrhosis.

According to our results, the vast majority of F3 patients obtaining SVR can be safely suspended from HCC surveillance if well defined at baseline for clinical, biochemical and ultrasonographic parameters.

A not negligible absolute HCC risk persists in cirrhotic patients even if cured from HCV infection, and more data are needed before establishing definitive algorithms able to drive cost-effective strategies of HCC surveillance on an individual basis.



Citation: Ciancio, A.; Ribaldone, D.G.; Spertino, M.; Risso, A.; Ferrarotti, D.; Caviglia, G.P.; Carucci, P.; Gaia, S.; Rolle, E.; Sacco, M.; et al. Who Should Not Be Surveilled for HCC Development after Successful Therapy with DAAS in Advanced Chronic Hepatitis C? Results of a Long-Term Prospective Study. *Biomedicines* 2023, *11*, 166. https://doi.org/10.3390/ biomedicines11010166

Academic Editor: Albrecht Piiper

Received: 2 November 2022 Revised: 12 December 2022 Accepted: 4 January 2023 Published: 9 January 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

2. Introduction

The clearance of Hepatitis C Virus (HCV) by Direct Acting Antiviral Agents (DAAs) reduces but does not eliminate the risk of developing Hepatocellular Carcinoma (HCC) in patients with advanced chronic liver disease (aCLD) [1]. For this reason, there is general agreement that cirrhotic patients should undergo HCC surveillance after achieving Sustained Virologic Response (SVR), but the international guidelines are divergent regarding the follow-up of F3 patients. According to AASLD guidelines [2], F3 patients obtaining SVR may be safely discharged, while the European guidelines [3] are more stringent, recommending ultrasonography every 6 months in both F3 and F4 patients, irrespective of the results of Non Invasive Tests (NITs) performed before and after SVR achievement [4]. This discrepancy is mainly based upon the difficulty in precisely defining patients with F3 fibrosis; moreover, the histologic stage is not the only predictor of HCC and for this reason there is currently a general consensus that attribution to the F3 stage should depend not only on histologic/elastographic aspects but also on ultrasonographic, clinical and biochemical variables [1,5] in order to accurately predict the HCC risk on an individual basis. To date, several studies [1,6–8] have tried to define the most effective NITs and related algorithms for identifying who to follow over time, while few studies with adequate follow-up [5,9–11] have addressed the problem of sustained responders to DAAs with baseline pre-cirrhotic liver fibrosis who can be safely suspended from long-term follow-up. This issue is relevant, as its definition would avoid unnecessary examinations and visits, freeing up resources for patients at high risk of developing HCC.

The primary aim of our study was to establish the long-term risk of HCC in patients with aCLD successfully treated by DAAs stratified according to their baseline measurement of liver stiffness (LSM), liver ultrasonographic aspect, clinical history, and Fibrosis-4 Index for Liver Fibrosis (FIB-4)/AST to Platelet Ratio Index (APRI) values. The secondary aim was to verify whether a post-SVR change in FIB-4/APRI values adds relevant information regarding patients who may terminate long-term HCC surveillance.

3. Patients and Methods

Due to Italian National Health System rules, since the introduction of DAAs in Italy from 2015 to 2017, only patients with advanced liver fibrosis (F3–F4 according to METAVIR Score [12]) could be reimbursed for treatment with DAAs; for this reason, all consecutive patients with HCV-positive aCLD referred to the Gastrohepatologic Clinic of Molinette Hospital, Turin, Italy, for DAAs therapy (sofosbuvir-based 87%) according to the EASL guidelines [13] between 1 January 2015 and 31 December 2017 were considered.

Inclusion criteria were as follows: >18 years; positive HCV-RNA by polymerase chain reaction (PCR); F3 and F4 liver fibrosis assessed by Transient Elastography (TE) within 3 months prior to inclusion into the study. Exclusion criteria were lack of written informed consent, TE not feasible or unreliable measurement, patients on waiting list for orthotopic liver transplant (OLT), post-OLT patients, past or current history of HCC, concomitant liver diseases such as haemochromatosis, Wilson's disease, drug-related liver disease, autoimmune hepatitis, HBsAg carriership, Human immunodeficiency Virus (HIV) infection, primary biliary cholangitis, and alpha-1-antitrypsin deficiency.

Patients were considered cirrhotic if showing LSM > 14.5 kPa by TE; stage 3 liver fibrosis was established by LSM values ranging from 9.5 to 14.5 kPa [14,15], but patients with a nodular liver surface, splenomegaly, porto-systemic collaterals detected by abdominal ultrasound (US), platelets < 120×10^9 /L, previous or current history of ascites, encephalopathy and variceal bleeding, or esophageal varices detected by endoscopy were considered cirrhotic [16], irrespective of their baseline LSM value. The reliability criteria for LSM were as follows: 10 valid measurements achieved with a success number $\geq 60\%$ and an interquartile range-to-median ratio $\leq 30\%$. A minimum of 3 h fasting was required.

Out of 1188 consecutive patients treated with DAAs, 146 (12.2%) were excluded (HBV and/or HIV carriers, past or current history of HCC, pre/post-OLT patients, TE not feasible, lack of informed consent). Of the remaining 1042 patients, 23 (2.2%) did not achieve SVR

(negative viremia 12 weeks after the end of treatment) and were excluded from the study. Out of 1019 sustained responders, 19 (1.9%) were lost to follow up. Therefore, the final analysis was performed in 1000 patients with SVR. The patient flow is reported in Figure 1.



Figure 1. Flowchart of recruited patients.

At baseline, a complete medical history and physical examination was undertaken and the following data were obtained from each patient: age, gender, ethnicity, smoking habits, alcohol intake, body mass index (BMI), waist circumference (WC), duration of HCV infection, and relevant co-morbidities. The following data on laboratory parameters were also recorded to define baseline characteristics: complete blood count, routine liver biochemistry (alanine aminotransferase [ALT] and AST, total bilirubin, albumin, alkaline phosphatase [APH], gamma glutamyltranspeptidase [GGT]), international normalized ratio [INR], creatinine, fasting plasma glucose [FPG], total cholesterol and HDL, triglycerides, HCV genotype, and viral load [AmpliPrep[®]/COBAS Taqman[®] HCV test, Roche Diagnostics, Basel, Switzerland]. We adopted an algorithm based on BMI, waist circumference, triglycerides and GGT to detect and graduate baseline liver steatosis by the fatty liver index [17]: a significant liver steatosis was associated with patients showing a cut-off ≥ 60 .

For each patient, we calculated FIB-4 [(Age (years)) × AST (U/L)/(Platelets $(10^9/L) \setminus 2\sqrt{ALT (U/L)}$] [18] and APRI [(AST/AST upper limit normal)/Platelets $(10^9/L) \times 100$] [19]

scores at baseline and at SVR achievement. An abdominal ultrasound (US) was performed on each patient at baseline in order to exclude HCC. Cirrhotic patients were stratified according to the Child-Turcotte-Pugh classification and Model for End-Stage Liver Disease [MELD] and underwent endoscopy to verify the presence of esophageal varices.

Patients were divided into 4 groups according to their baseline characteristics:

- (1) Group 1: patients with LSM $\ge 9.5 \le 14.5$ kPa and FIB-4 < 3.25 and APRI < 1.5
- (2) Group 2: patients with LSM $\ge 9.5 \le 14.5$ kPa and FIB-4 ≥ 3.25 or APRI ≥ 1.5
- (3) Group 3: patients with LSM > 14.5 kPa or clinical/biochemical/US signs of cirrhosis and FIB-4 < 3.25 and APRI < 1.5</p>
- (4) Group 4: patients with LSM > 14.5 kPa or clinical/biochemical/US signs of cirrhosis and FIB-4 ≥ 3.25 or APRI ≥ 1.5

We used the FIB-4 cut-off of \geq 3.25 due to its positive predictive value of 82.1% for advanced fibrosis [20]; in order to better classify patients showing an indeterminate range (1.45–3.25), we used the APRI cut-off \geq 1.5 to individuate carriers of significant fibrosis [21].

An abdominal US was planned at 6-month intervals and patients were followed until OLT, death or until the end of December 2021. Patients were also censored at the moment of HCC diagnosis, performed according to international guidelines [22,23] by histological examination or by contrast-enhanced imaging methods showing hypervascularity in late arterial phase and washout on portal venous and/or delayed phases. They were followed-up for a median of 48 (IQR: 36–60) months after achieving SVR.

The study was performed in accordance with the principles of the Declaration of Helsinki and approved by the local ethics committee (Comitato Etico Interaziendale Città della Salute e della Scienza di Torino, Turin, Italy, n 452), and written informed consent was obtained from all patients.

Statistical Analysis

Continuous variables were reported as median (inter-quartile range (IQR)). Normality was checked by the D'Agostino–Pearson test. Categorical variables were reported as number and percentage. Comparison of continuous variables between independent groups was performed by the Mann–Whitney test. Regarding the dichotomous categorical variable, a Chi-squared test was performed for unpaired analysis. Comparison of Kaplan-Meier survival curves was performed using the Logrank test. The association between variables was assessed by Cox proportional hazards, the strength of association was reported as hazards ratio (H.R.) and 95% CI, respectively.

All statistical analyses were performed using MedCalc[®] v.18.9.1 (MedCalc Software Ltd., Ostend, Belgium), and a p value ≤ 0.05 was considered statistically significant.

4. Results

Baseline characteristics of the patients divided into four groups are reported in Tables 1 and 2.

| Characteristics | Group 1 <i>n</i> = 324 (32.4%) | Group 2 <i>n</i> = 133 (13.3%) | Group 3 <i>n</i> = 158 (15.8%) | Group 4 <i>n</i> = 385 (38.5%) | р |
|--|-----------------------------------|-----------------------------------|-----------------------------------|-----------------------------------|---|
| Age (years), median (IQR) | 56 (50–65) | 68 (60–75) | 53 (48–64) | 65 (55–74) | Group 1 vs. 2/4 = 0.0001 Group 2 vs. 3/4 = 0.002 Group 3 vs. 4 = 0.0001 |
| Males, <i>n</i> (%) Females, <i>n</i> (%) | 186 (57.4%) 138 (42.6%) | 60 (45.1%) 73 (54.5%) | 112 (70.9%) 46 (29.1%) | 221 (57.4%) 164 (42.6%) | Group 1 vs. 2/3 = 0.002 Group 2 vs. 3/4 = 0.01 Group 3 vs. 4 = 0.003 |

Table 1. Baseline demographic and clinical characteristics of 1000 sustained responders to DAAs.

| Characteristics | Group 1 <i>n</i> = 324 (32.4%) | Group 2 <i>n</i> = 133 (13.3%) | Group 3 <i>n</i> = 158 (15.8%) | Group 4 <i>n</i> = 385 (38.5%) | р |
|--|-----------------------------------|-----------------------------------|-----------------------------------|-----------------------------------|--|
| Ethnicity Caucasian, n (%) African, n (%) | 316 (97.5%) 8 (2.5%) | 133 (100%) 0 (0%) | 155 (98.1%) 3 (1.9%) | 381 (99%) 4 (1%) | N.S. |
| BMI, median (IQR) | 24.8 (22.6–27.4) | 24.2 (22.1–27.8) | 25 (22.8–28.4) | 24.7 (22.7–27.1) | N.S. |
| BMI > 25, n (%) | 153 (47.2%) | 54 (40.6%) | 79 (50%) | 178 (46.2%) | N.S. |
| Abnormal waist circumference, <i>n</i> (%) | 178 (54.9%) | 94 (70.6%) | 86 (54.4%) | 253 (65.7%) | Group 1 vs. 2/4 = 0.002 Group 2 vs. 3 = 0.005 Group 3 vs. 4 = 0.01 |
| Obesity, n (%) | 35 (10.8%) | 18 (13.5%) | 19 (12.0%) | 38 (9.9%) | N.S. |
| Duration of infection (years), median (IQR) | 17 (10–23) | 19 (11–24) | 18 (7–24) | 20 (11–24) | Group 1 vs. 4 = 0.01 |
| Smoking status Never, <i>n</i> (%) Past or current, <i>n</i> (%) | 291 (89.8%) 33 (10.2%) | 121 (91%) 12 (9%) | 130 (82.3 %) 28 (17.7%) | 345 (89.6%) 40 (10.4%) | N.S. |
| Alcohol intake No, <i>n</i> (%) Yes, <i>n</i> (%) | 315 (97.2%) 9 (2.8%) | 132 (99.2%) 1 (0.8%) | 149 (94.3%) 9 (5.7%) | 375 (97.4%) 10 (2.6%) | N.S. |
| Diabetes, n (%) | 35 (10.8%) | 23 (17.2%) | 35 (22.2%) | 82 (21.3%) | Group 1 vs. 3/4 = 0.0002 |
| Metabolic syndrome, n (%) | 47 (14.5%) | 24 (18%) | 32 (20.3%) | 73 (19.0%) | N.S. |
| Baseline co-morbidities, <i>n</i> (%) | 67 (20.7%) | 48 (36.1%) | 44 (27.8%) | 115 (29.9%) | Group 1 vs. 2/4 = 0.005 |
| Fatty liver index, median (IQR) | 37.2 (21.4–57.9) | 46.2 (26.8–63.8) | 48.8 (28.3–73.0) | 47.6 (30.5–68.0) | Group 1 vs. 2/3/4 = 0.0001 |
| Fatty liver index \geq 60, <i>n</i> (%) | 77 (23.8%) | 37 (27.8%) | 65 (41.1%) | 135 (35.1%) | Group 1 vs. 3/4 = 0.001 Group 2 vs. 3 = 0.02 |
| Esophageal varices, n (%) | 0 (0%) | 0 (0%) | 26 (16.5%) | 178 (46.2%) | Group 1/2 vs. 3/4 = 0.0001 |

 Table 1. Cont.

DAAs, Direct Antiviral Agents; IQR, Interquartile range; BMI, Body Mass Index; N.S., Not Significant.

Table 2. Baseline biochemical, virologic and liver-related characteristics of 1000 sustained responders to DAAs.

| Characteristics | Group 1 <i>n</i> = 324 (32.4%) | Group 2 <i>n</i> = 133 (13.3%) | Group 3 <i>n</i> = 158 (15.8%) | Group 4 <i>n</i> = 385 (38.5%) | р |
|--|-----------------------------------|-----------------------------------|-----------------------------------|-----------------------------------|--|
| Liver stiffness (kPa), median (IQR) | 11.6 (10.4–12.8) | 12.6 (11.4–14.0) | 18.2 (16.0–23.7) | 23.4 (16.9–32.4) | Group 1 vs. 2/3/4 = 0.0001 Group 2 vs. 3 = 0.0001 Group 3 vs. 4 = 0.0001 |
| FIB-4 score, median (IQR) | 1.8 (1.3–2.4) | 4.5 (3.7–6.1) | 2.1 (1.5–2.7) | 6.2 (4.5–9.3) | Group 1 vs. 2/3/4 = 0.0001 Group 2 vs. 3/4 = 0.003 Group 3 vs. 4= 0.0001 |
| APRI score, median (IQR) | 0.3 (0.2–0.4) | 0.7 (0.4–1.1) | 0.3 (0.2–0.4) | 0.9 (0.6–1.4) | Group 1 vs. 2/3/4 = 0.003 Group 2 vs. 3/4 = 0.001 Group 3 vs. 4 = 0.0001 |
| MELD score, median (IQR) | 7 (6–7) | 7 (7–9) | 7 (7–8) | 8 (7–10) | Group 1 vs. 2/3/4 = 0.003 Group 2 vs. 3/4 = 0.03 Group 3 vs. 4= 0.0001 |
| MELD > 15, <i>n</i> (%) | 0 (0%) | 0 (0%) | 8 (5.1%) | 16 (4.2%) | Group 1 vs. 3/4 = 0.0001 Group 2 vs. 3/4 = 0.0001 |
| AST (IU/mL), median (IQR) | 47 (33.5–57.5) | 80 (50–121.8) | 50 (33–68) | 84 (57–117) | Group 1 vs. 2/4 = 0.0001 Group 2 vs. 3 = 0.001 Group 3 vs. 4 = 0.0001 |
| ALT (IU/mL), median (IQR) | 64 (41–87) | 79 (51.8–140.5) | 64.5 (38-88) | 72 (51–121) | Group 1 vs. 2/4 = 0.0001 Group 2 vs. 3 = 0.0001 Group 3 vs. 4 = 0.0002 |

| Characteristics | Group 1 <i>n</i> = 324 (32.4%) | Group 2 <i>n</i> = 133 (13.3%) | Group 3 <i>n</i> = 158 (15.8%) | Group 4 <i>n</i> = 385 (38.5%) | р |
|--|--|--|--|---|--|
| GGT (IU/mL), median (IQR) | 31 (28–71) | 60 (37.8–102.3) | 64 (51–105) | 64 (48–103) | Group 1 vs. 2/3/4 = 0.0008 Group 2 vs. 4 = 0.02 |
| Platelets count (×10 ³ /mm ³), median (IQR) | 188 (159–231) | 123 (98–150) | 163 (133–213) | 92 (66–123) | Group 1 vs. 2/3/4 = 0.0002 Group 2 vs. 3/4 = 0.0001 Group 3 vs. 4 = 0.0001 |
| Albumin (g/dL), median (IQR) | 4.4 (4.2–4.6) | 4.2 (3.9–4.4) | 4.3 (4-4.5) | 3.9 (3.5–4.2) | Group 1 vs. 2/3/4 = 0.007 Group 2 vs. 3/4 = 0.0007 Group 3 vs. 4 = 0.0001 |
| Total cholesterol (mg/dL), median (IQR) | 158 (131–183) | 149 (130–170) | 144 (130 - 165) | 130 (123–156) | Group 1 vs. 2/3/4 = 0.005 Group 2 vs. 4 = 0.0003 Group 3 vs. 4 = 0.0001 |
| Triglycerides (mg/dL), median (IQR) | 98 (72–116) | 96 (75–109) | 96 (68–110) | 100 (77–107) | N.S. |
| Genotypes, 1 2 3 4 | 236 (72.8%) 15 (4.6%) 35 (10.8%) 38 (11.8%) | 106 (79.7%) 8 (6.0%) 6 (4.5%) 13 (9.8%) | 113 (71.5%) 6 (3.8%) 15 (9.5%) 24 (15.2%) | 290 (75.3%) 18 (4.7%) 30 (7.8%) 47 (12.2%) | N.S. |

Table 2. Cont.

DAAs, Direct Antiviral Agents; IQR, Interquartile range; N.S., Not Significant.

Group 1 patients showed lower median LSM (11.6 kPa [IQR: 10.4–12.8]) compared with the other groups (p = 0.0001), a shorter duration of infection (median 17 years, [IQR: 10–23], p = 0.01), fewer baseline co-morbidities (20.7%, p = 0.005), a lower rate of diabetes mellitus (10.8%, *p* = 0.0002), and a significantly lower fatty liver index (37.2 [IQR: 21.4–57.9], *p* = 0.0001). Markers of liver synthesis were significantly higher in group 1 patients (median albumin value: 4.4 g/dL [IQR:4.2–4.6], *p* = 0.007, median total cholesterol level: 157.5 mg/dL [IQR: 130.5–182.5], p = 0.005) as well as the median platelet count (187.5 \times 10³/mm³ [IQR: 158.5–231.0], p = 0.0002), confirming a less advanced liver disease (median MELD: 7.0 [IQR: 6.0-7.0], p = 0.0001) and a lower stage of liver fibrosis (median FIB-4 score: 1.8 [IQR: 1.3-2.4], p = 0.0001, median APRI score: 0.3 [IQR: 0.2–0.4], p = 0.003). Overall, there were 71 incident cases of HCC diagnosed during follow-up, with an HCC incidence rate (IR) of 1.97/100 patients/year (PY) and a cumulative incidence rate (CIR) of 2.2%, 5.5% and 9.5% at 12, 36 and 60 months, respectively. In Group 1, one out of 324 patients (0.3%) developed HCC 7 months after SVR achievement, corresponding to an HCC IR of 0.09/100 PY and a CIR at 12, 36 and 60 months of 0.3%, compared with 6/133 (4.5%) in Group 2 (1.22/100 PY, CIR of 2.3%, 3.3%, 5.7%), 10/158 (6.3%) in Group 3 (1.68/100 PY, CIR of 0.7%, 4.2%, 8.6%) and 54/385 (14.0%) in Group 4 (4.01/100 PY, CIR of 4.3%, 11.3%, 17.8%). The difference in HCC incidence between Group 1 and the other three Groups was statistically significant (p = 0.0009 vs. Group 2, p = 0.0001 vs. Group 3, p < 0.0001 vs. Group 4); a significant lower HCC IR was also observed in Group 2 compared with Group 4 (p = 0.004) and in Group 3 vs. Group 4 (p = 0.009). (Figure 2). In order to individuate independent predictors of HCC occurrence, we checked patients for each baseline demographic, clinical and biochemical characteristic reported in Tables 1 and 2; Table 3 presents the factors associated with the long-term risk of HCC by univariate and multivariate analysis. Pre-therapy variables which were not significant in the models were not reported. In particular, the prevalence of various genotypes did not significantly change among patients who developed (or did not) HCC (genotype 1 = 71.7% vs. 69.5%, genotype 2 = 1.6% vs. 4.6%, genotype 3 = 8.4% vs. 8.6%, genotype 4 = 18.3% vs. 11%, p = 0.43). According to the univariate analysis, many characteristics were associated with HCC occurrence (older age, male gender, high LSM, cirrhosis, presence of esophageal varices, APRI/FIB-4/MELD scores, low levels of albumin, cholesterol and platelet count, belonging to group 2–4) but only male gender (HR = 1.93; 95% CI = 1.12-3.31, p = 0.02), cirrhosis (HR = 3.6; CI = 2.42-5.88, p = 0.001), the presence of esophageal varices (HR = 1.76; CI = 1.03-3.01, p = 0.04), and belonging to group 2-4 (HR = 10.95; CI = 1.19-100.51, p = 0.03) were independent predictors of HCC. At the end of the follow-up, there was an overall mortality of 60 out of 1000 patients (6%): 7/324 (2.1%) in group 1, 6/133 (4.5%) in group 2, 12/158 (7.6%) in

group 3, and 35/385 (9.1%) in Group 4 (Figure 3). Death occurred due to liver-related causes in 42 patients (70%): one out of 42 (2.4%) belonged to Group 1 and died for the HCC occurrence as well as 3 Group 2 patients (7.1%). The remaining 18 patients died due to extrahepatic neoplastic causes or cardiovascular diseases. Ten patients (1.0%) were referred for liver transplantation, 2 of them (20%) belonging to Group 2 and 8 (80%) to Group 4. The difference in terms of mortality was statistically significant between Group 1 and Group 3 (p = 0.007) and between Group 1 and Group 4 (p = 0.0001).

Impact of Change in FIB-4/APRI on HCC Risk

Due to the very small proportion (0.8%) of patients showing an increase in FIB-4/APRI scores after achieving SVR, we focused on Group 2 and 4 patients and we observed that the risk of incident HCC was reduced among patients who experienced a decrease of FIB-4/APRI scores below 3.25 and 1.5, respectively. In Group 2, two out of 85 (2.4%) patients showing a decline in FIB-4/APRI developed HCC compared with 4/48 (8.3%) patients with scores above the cut-offs, corresponding to an IR of 0.6/100 PY vs. 2.8/100 PY and a CIR at 12, 36 and 60 months of 1,2%, 2.6% and 2.6%, respectively, compared to 4.7%, 4.7% and 14.3%, p = 0.049 (Figure 4), respectively. In Group 4, 12/167 (7.2%) patients with a significant decrease in FIB-4/APRI scores showed "de novo" HCC compared to 42/218 (19.3%) patients with no relevant changes; HCC IR was 1.7/100 PY in the first sub-group versus 6.4/100 PY in the second one, with a CIR of 0.6%, 5.8% and 8.1% compared with 7.2%, 15.9%, 26.8%, respectively, p < 0.0001 (Figure 5). Baseline characteristics of patients stratified according to their improvement in FIB-4/APRI scores are reported in Table 4; the only independent predictor of HCC by multivariate analysis was the lack of FIB-4/APRI improvement (H.R. = 0.02 [95% CI: 0.004–0.15]).



Figure 2. Kaplan-Meier HCC-free survival curves of patients according to group.

| | Hazard Ratio (95% CI) | | | |
|--|------------------------|----------|------------------------|-------|
| Characteristic | Univariate | р | Multivariate | р |
| Age in years | 1.02 (1.00–1.05) | 0.03 | 1.02 (0.996–1.043) | 0.10 |
| Male gender | 1.75 (1.05–2.91) | 0.03 | 1.93 (1.12–3.31) | 0.02 |
| Liver stiffness (kPa) | 1.04 (1.03–1.06) | <0.0001 | 1.01 (0.997–1.030) | 0.12 |
| Cirrhosis (yes) | 4.02 (2.81–6.43) | 0.0001 | 3.6 (2.42–5.88) | 0.001 |
| APRI > 1.5 | 3.42 (2.04–5.75) | < 0.0001 | 1.36 (0.76–2.44) | 0.31 |
| FIB-4 > 3.25 | 4.34 (2.38–7.99) | < 0.0001 | 0.82 (0.35–1.92) | 0.65 |
| MELD > 15 | 4.01 (1.61–9.58) | 0.003 | 2.3 (0.88–6.00) | 0.09 |
| Esophageal varices (yes) | 4.52 (2.83–7.20) | < 0.0001 | 1.76 (1.03–3.01) | 0.04 |
| Platelet count < 120.000/mm ³ | 4.28 (2.55–7.18) | < 0.0001 | 1.54 (0.75–3.18) | 0.24 |
| Albumin < 4 g/dL | 1.68 (1.55–1.75) | < 0.0001 | 1.37 (0.97–1.61) | 0.06 |
| Total cholesterol < 150 mg/dL | 1.01 (1.002–1.16) | 0.04 | 1.06 (0.998–1.014) | 0.17 |
| Group 2 (reference Group 1) | 14.49 (1.74–120.34) | 0.01 | 10.95 (1.19–100.51) | 0.03 |
| Group 3–4 (reference Group 1) | 38.83 (5.39–279.91) | 0.0003 | 15.8 (2.03–123.44) | 0.009 |

Table 3. Association between baseline characteristics and development of HCC.

HCC, Hepatocellular Carcinoma; CI, confidence interval.



Figure 3. Kaplan-Meier overall-survival curves of patients according to group.



Figure 4. Kaplan-Meier HCC-free survival curves of Group 2 patients according to changes in FIB-4/APRI scores.

Table 4. Baseline characteristics of Group 2 and 4 patients according to their improvement in FIB-4/APRI scores.

| | Improved (252 pts) | Not Improved (266 pts) | р |
|---|---------------------------------|---------------------------------|-----------------|
| Age (years), median [IQR] | 66 [57–74] | 66.5 [56–74] | 0.89 |
| Males (<i>n</i> , %) Females (<i>n</i> , %) | 124 (49.2%) 128 (50.8%) | 157 (59%) 109 (41%) | 0.03 |
| Ethnicity Caucasian (<i>n</i> , %) African (<i>n</i> , %) | 252 (100%) 0 (0%) | 262 (98.5%) 4 (1.5%) | 0.051 |
| BMI, median (IQR) BMI > 25, <i>n</i> (%) | 24.1 (22.1–26.9) 158 (37.3%) | 25.3 (22.9–27.6) 138 (51.9%) | 0.004 0.0009 |
| Abnormal waist circumference, n (%) | 177 (70.2%) | 170 (63.9%) | 0.62 |
| Obese patients $(n, \%)$ | 26 (10.3%) | 30 (11.3%) | 0.73 |
| Smoking status Never (n, %) Past or current smokers (n, %) | 230 (91.3%) 22 (8.7%) | 236 (88.7%) 30 (11.3%) | 0.16 |
| Alcohol abuse No (<i>n</i> , %) Yes (<i>n</i> , %) | 245 (97.2%) 7 (2.8%) | 262 (98.5%) 4 (1.5%) | 0.32 |

| | Improved (252 pts) | Not Improved (266 pts) | р |
|---|-----------------------|---------------------------|----------|
| Infection duration (years), median (IQR) | 20 (11–24) | 19 (13–24) | 0.76 |
| Metabolic syndrome (<i>n</i> , %) | 57 (22.6%) | 40 (15%) | 0.03 |
| Baseline co-morbidities $(n, \%)$ | 78 (31%) | 85 (32%) | 0.81 |
| Liver stiffness (kPa), median (IQR) | 17.3 (14–26.8) | 20.6 (14–30.7) | 0.009 |
| FIB-4, median (IQR) | 4.5 (3.8–5.8) | 7.7 (4.0–11.5) | < 0.0001 |
| APRI, median (IQR) | 0.7 (0.5–1) | 1 (0.7–1.6) | < 0.0001 |
| MELD > 15, <i>n</i> (%) | 3 (1.2%) | 13 (4.9%) | 0.02 |
| Esophageal varices, n (%) | 51 (20.2%) | 127 (47.7%) | < 0.0001 |
| ALT (IU/L), median (IQR) | 81 (55.5–140) | 70 (48–109) | 0.0004 |
| GGT (IU/L), median (IQR) | 62 (45–109) | 61.5 (46–100) | 0.6 |
| Platelet count ($\times 10^3$ /mm ³), median (IQR) | 120 (99–145) | 80.5 (57–106) | < 0.0001 |
| Albumin (g/dL), median (IQR) | 4.2 (3.9–4.4) | 3.8 (3.4–4.1) | < 0.0001 |
| Total cholesterol (mg/dL), median (IQR) | 144 (130–165.5) | 130 (120–155) | 0.0004 |
| FLI, median (IQR) | 46.6 (28–68.3) | 47.2 (30.3–67.4) | 0.72 |

Table 4. Cont.

BMI, body mass index; MELD, Model of End Stage Liver Disease; IQR, Inter Quartile Range; ALT, alanine aminotransferase; GGT, gamma glutamyl transpeptidase; FLI, Fatty Liver Index.



Figure 5. Kaplan-Meier HCC-free survival curves in Group 4 patients according to changes in FIB-4/APRI scores.

5. Discussion

The current International Guidelines [2,3] diverge on the surveillance strategy regarding patients with aCLD achieving SVR, in particular those ones with baseline F3 fibrosis. The EASL [3] recommends long-term US surveillance in F3 patients, whereas AASLD does not [2]. This divergence is mainly based upon the difficulty in establishing an accurate diagnosis of F3 fibrosis: liver biopsy can underestimate the presence of cirrhosis in patients histologically labeled as F3 [24], while TE is unable to distinguish F3 fibrosis from cirrhosis with certainty [25]. In order to better stratify the neoplastic risk in this particular subset of patients, recent studies [9–11] have tried to associate algorithms composed by demographic, clinical, biochemical risk factors to the histological or elastographic assessment, but to date none of them have been officially recommended.

According to our data, the HCC incidence in patients with LSM $\geq 9.5 \leq 14.5$ kPa achieving SVR is negligible (0.09/100 PY) in the vast majority of them after a median follow-up of 4 years; the identification of this low-risk group can be performed easily before treatment by associating clinical, biochemical variables and US imaging to LSM. Moreover, a significant decline in FIB-4/APRI scores after SVR achievement in this particular subset of patients with high baseline values is associated with a decreased risk of HCC (0.6/100 PY), enabling the individuation of additional patients at low risk of HCC. Overall, 90% of patients with LSM $\geq 9.5 \leq 14.5$ kPa were identified as having an HCC risk < 1%/year, well below the cost-effectiveness cut-off for HCC surveillance. On the other hand, the remaining 10% of patients with persistently high FIB-4/APRI scores were probably carriers of undiagnosed cirrhosis, therefore needing an HCC surveillance program due to the relevant HCC incidence found (2.8/100 PY).

Our results are similar to those reported by recent studies [5,9–11], which specifically addressed the issue of HCC risk in pre-cirrhotic patients. By adopting algorithms based upon FIB-4/APRI scores, Kanwal et al. [10] showed that only 4% of patients without a diagnosis of cirrhosis at baseline require HCC surveillance. Sanchez-Azofra et al. [5] followed-up a large cohort of patients with baseline HCV-positive stage-3 fibrosis successfully treated by DAAs, finding a low HCC incidence rate (0.47/100 PY), suggesting the maintenance of surveillance only in males older than 55 years of age. According to our data, male sex is an independent predictor of HCC, as well as cirrhosis, persistently high FIB-4/APRI scores, and clinical significant portal hypertension, confirming what was reported by Azzi et al. [9]. Surprisingly, baseline co-morbidities and variables such as diabetes, metabolic syndrome, high BMI, visceral obesity, and fatty liver index were not selected as independent HCC risk predictors, but this finding is consistent with data from previous studies [5,9–11]. In particular, diabetes mellitus did not emerge as an unfavourable independent predictive factor for HCC risk upon multivariate analysis. Data from follow-up studies on patients with SVR after therapy with DAAs [9–11,26–38] are conflicting, as diabetes was not found to be an independent predictor of HCC in most of them [9-11,26-28,30-35].

Our results confirmed what was already found by Kanwal et al. [10], who showed that the HCC risk remained above the accepted threshold for surveillance in cirrhotic patients, even in the case of significant decline in FIB-4/APRI scores after therapy. However, at variance with our data, a recent study [11] showed that a post-treatment algorithm combining age/albumin/TE and, optionally, alfafetoprotein and alcohol intake enables a reliable risk stratification for the development of HCC, even among cirrhotic patients, identifying a large proportion of patients with an HCC risk < 1%/year, below the threshold considered to be cost-effective. Cost-effectiveness analyses regarding HCC surveillance changed over time; Cucchetti et al. [39] quoted an annual HCC risk threshold > 1.5%, above which HCC screening was cost-effective, but recent studies [40,41] focusing on patients achieving SVR suggested lower thresholds, from 1.32/100 PY up to 0.5/100 PY. According to Mueller et al. [42], HCC surveillance is cost-effective until the age of 70 for cirrhotic patients cured of hepatitis C and until the age of 60 for patients with non-cirrhotic advanced fibrosis. Irrespective of the thresholds considered, we acknowledge that FIB-4/APRI-based algorithms may not be the best method to stratify the HCC risk in cirrhotic patients achieving SVR; we have recently compared [43] different non-invasive scoring systems (the Forns index, FIB-4, albumin-bilirubin score [ALBI], age/gender/albumin-bilirubin/platelets score [aMAP]) in cirrhotics successfully treated by DAAs, and we found that ALBI showed the highest diagnostic accuracy for the detection of HCC, confirming previous data [44–46]. However, given the controversies regarding the markers and algorithms which should be adopted, we think that results from large prospective studies with longer follow-up are still needed before incorporating definite algorithms into official recommendations regarding the optimal HCC surveillance strategy in a cirrhotic population cured from HCV infection.

Very similar to what was previously reported [47], death occurred in about 6% of our patients due to liver-related causes in the majority of them (70%); however, the mortality rate was significantly lower among Group 1 patients compared with Group 3–4 patients, confirming the excellent clinical outcome of F3 patients with low FIB-4/APRI scores obtaining SVR. The strengths of our study are the prospective design, the large sample size with long-term follow-up, and the very low rate of patients lost to follow-up. However, our study was conducted in a single academic centre mainly recruiting Caucasian patients without a validation cohort, therefore limiting the reproducibility of our conclusions, and multicenter prospective studies are needed in order to make a solid recommendation.

Moreover, we acknowledge the possibility that some cirrhotic patients could have been included in Group 1 and 2, due to the wide range of kPa adopted to define F3 patients and despite the accurate clinical and ultrasonographic screening, however, according to our results, our approach would still work for these patients.

In conclusion, the great majority of patients with LSM $\ge 9.5 \le 14.5$ kPa achieving SVR can be safely suspended from HCC surveillance if well defined at baseline for clinical, biochemical and US parameters, which are all currently used, before starting antiviral therapy with no cost increases. On the other hand, a word of caution is still necessary regarding the withdrawal or deferral of HCC surveillance in cirrhotic patients cured from HCV infection, as a late diagnosis of HCC implies severe consequences for an individual patient.

Author Contributions: Conceptualization, G.M.S., A.C.; methodology, A.C., G.M.S., D.G.R.; software, M.S. (Marco Sacco), D.G.R.; validation, D.G.R. and M.S. (Marco Sacco); formal analysis, D.G.R.; investigation, A.C., D.G.R., M.S. (Marco Sacco), M.S. (Matteo Spertino), A.R., D.F., G.P.C., P.C., S.G., E.R.; data curation, D.G.R., M.S. (Marco Sacco), M.S. (Matteo Spertino) and G.M.S.; writing—original draft preparation, G.M.S.; writing—review and editing, D.G.R. and M.S. (Marco Sacco); visualization, M.S. (Marco Sacco); supervision, G.M.S.; project administration, G.M.S. All authors have read and agreed to the published version of the manuscript.

Funding: The project has been partially supported by a Gilead Sciences Medical Grant.

Institutional Review Board Statement: The study was conducted in accordance with the Declaration of Helsinki, and approved by the Institutional Ethics Committee of the Città della Salute e della Scienza—University Hospital of Turin, Italy (2CEI/452).

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

Data Availability Statement: The data that support the findings of this study are available from the corresponding author upon reasonable request.

Conflicts of Interest: The authors declare that they have no conflict of interest.

Abbreviation

HCV, hepatitis C virus; DAAs, direct-acting antiviral agents; HCC, Hepatocellular Carcinoma; SVR, sustained virologic response; TE, transient elastography; FIB-4, Fibrosi-4 Index for Liver Fibrosis; APRI, aspartate aminotransferase to platelet ratio; OLT, orthotopic liver transplantation; HIV, Human Immunodeficiency Virus; EASL, European Association for the Study of the Liver; AASLD, American Association for the Study of the Liver; BMI, body mass index; WC, waist circumference; DM, type 2 diabetes mellitus; ALT, alanine aminotransferase; APH, alkaline phosphatase; GGT, gamma glutamyltranspeptidase; US, abdominal ultrasound; MELD, model for end-stage liver diseases; IQR, Inter Quartile Range; CI, confidence interval; HR, hazard ratio.

References

- 1. Ioannou, G.N. HCC surveillance after SVR in patients with F3/F4 fibrosis. J. Hepatol. 2021, 74, 458–465. [CrossRef] [PubMed]
- Ghany, M.G.; Morgan, T.R. Panel A-IHCG. Hepatitis C guidance 2019 update: American Association for the Study of Liver Diseases-Infectious Diseases Society of America recommendations for testing, managing and treating hepatitis C virus infection. *Hepatology* 2020, 71, 686–721. [CrossRef] [PubMed]
- 3. European Association for the study of the Liver. EASL recommendations on treatment of Hepatitis C. Final update of the series. *J. Hepatol.* **2020**, *73*, 1170–1218. [CrossRef] [PubMed]
- 4. European Association for the Study of the Liver. Clinical practice guidelines on non-invasive tests for evaluation of liver disease severity and prognosis—2021 update. *J. Hepatol.* **2021**, *75*, 659–689. [CrossRef]
- Sanchez-Azofra, M.; Fernandez, I.; Garcia-Buey, M.L.; Dominguez-Dominguez, L.; Fernandez-Rodriguez, C.M.; Mancebo, A.; Bonet, L.; Ryan, P.; Gea, F.; Díaz-Sánchez, A.; et al. Hepatocellular Carcinoma risk in hepatitis C stage-3 fibrosis after sustained virological response with direct-acting antivirals. *Liver Int.* 2021, *41*, 2885–2891. [CrossRef] [PubMed]
- 6. Negro, F. Residual risk of liver disease after hepatitis C virus eradication. J. Hepatol. 2021, 74, 952–963. [CrossRef]
- Ahumada, A.; Rayon, L.; Uson, C.; Banares, R.; Alonso Lopez, S. Hepatocellular Carcinoma risk after viral response in hepatitis C virus-advanced fibrosis: Who to screen and for how long? *World J. Gastroenterol.* 2021, 27, 6737–6749. [CrossRef]
- 8. D'Ambrosio, R.; Degasperi, E.; Lampertico, P. Predicting Hepatocellular Carcinoma risk in patients with chronic HCV infection and a sustained virological response to direct-acting antivirals. *J. Hepatocell. Carcinoma* **2021**, *8*, 713–739. [CrossRef]
- Azzi, J.; Dorival, C.; Cagnot, C.; Fontaine, H.; Lusivika-Nzinga, C.; Leroy, V.; De Ledinghen, V.; Tran, A.; Zoulim, F.; Alric, L.; et al. Prediction of Hepatocellular Carcinoma in hepatitis C patients with advanced fibrosis after sustained virologic response. *Clin. Res. Hepatol. Gastroenterol.* 2022, 46, 101923. [CrossRef]
- 10. Kanwal, F.; Kramer, J.R.; Aseh, S.M.; Cao, Y.; Li, L.; El-Serag, H.B. Long-term risk of Hepatocellular Carcinoma in HCV patients treated with direct acting antiviral agents. *Hepatology* **2020**, *71*, 44–55. [CrossRef]
- Semmler, G.; Meyer, E.L.; Kozbial, K.; Schwabl, P.; Hametner-Schreil, S.; Zanetto, A.; Bauer, D.; Chromy, D.; Simbrunner, B.; Scheiner, B.; et al. HCC risk stratification after cure of hepatitis C in patients with compensated advanced chronic liver disease. *J. Hepatol.* 2022, *76*, 812–821. [CrossRef] [PubMed]
- 12. METAVIR Cooperative Study Group. Intraobserver and interobserver variations in liver biopsy interpretation in patients with chronic hepatitis C. *Hepatology* **1994**, *20*, 15–20. [CrossRef]
- 13. Pawlotsky, J.M.; Aghemo, A.; Back, D.; Dusheiko, G.; Forns, X.; Puoti, M.; Sarrazin, C. EASL recommendations on treatment of Hepatitis C 2015. *J. Hepatol.* **2015**, *63*, 199–236.
- 14. EASL-ALEH Clinical Practice Guidelines. Non-invasive tests for evaluation of liver disease severity and prognosis. *J. Hepatol.* **2015**, *63*, 237–264. [CrossRef]
- 15. Tsochatzis, E.A.; Gurusamy, K.S.; Ntaoula, S.; Cholongitas, E.; Davidson, B.R.; Burroughs, A.K. Elastography for the diagnosis of severity of fibrosis in chronic liver disease: A meta-analysis of diagnostic accuracy. *J. Hepatol.* **2011**, *54*, 650–659. [CrossRef]
- 16. Gaia, S.; Campion, D.; Evangelista, A.; Spandre, M.; Cosso, L.; Brunello, F.; Ciccone, G.; Bugianesi, E.; Rizzetto, M. Non-invasive score system for fibrosis in chronic hepatitis: Proposal for a model based on biochemical, FibroScan and ultrasound data. *Liver Int.* **2015**, *35*, 2027–2035. [CrossRef]
- 17. Bedogni, G.; Bellentani, S.; Miglioli, L.; Masutti, F.; Passalacqua, M.; Castiglione, A.; Tiribelli, C. The fatty liver index: A simple and accurate predictor of hepatic steatosis in the general population. *BMC Gastroenterol.* **2006**, *6*, 33. [CrossRef]
- Sterling, R.K.; Lissen, E.; Clumeck, N.; Sola, R.; Correa, M.C.; Montaner, J.; Sulkowski, M.S.; Torriani, F.J.; Dieterich, D.T.; Thomaset, D.L.; et al. Development of a simple noninvasive index to predict significant fibrosis in patients with HIV/HCV co-infection. *Hepatology* 2006, 43, 1317–1325. [CrossRef]
- 19. Wai, C.T.; Greenson, J.K.; Fontana, R.J.; Kalbfleisch, J.D.; Marrero, J.A.; Conjeevaram, H.S.; Lok, A.S.-F. A simple noninvasive index can predict both significant fibrosis and cirrhosis in patients with chronic hepatitis C. *Hepatology* **2003**, *38*, 518–526. [CrossRef]
- 20. Chou, R.; Wasson, N. Blood tests to diagnose fibrosis or cirrhosis in patients with chronic hepatitis C virus infection: A systematic review. *Ann. Intern. Med.* **2013**, *158*, 807–820. [CrossRef]
- Poynard, T.; McHutchison, J.; Manns, M.; Trepo, C.; Lindsay, K.; Goodman, Z.; Ling, M.; Albrecht, J. Impact of pegylated interferon alfa-2b and ribavirin on liver fibrosis in patients with chronic hepatitis C. *Gastroenterology* 2002, 122, 1303–1313. [CrossRef] [PubMed]
- 22. EASL Clinical Practice Guidelines. Management of Hepatocellular Carcinoma. J. Hepatol. 2018, 69, 182–236. [CrossRef] [PubMed]
- Singal, A.G.; Lim, J.K.; Kanwal, F. AGA clinical practice update on interaction between oral direct-acting antivirals for chronic hepatitis C infection and Hepatocellular Carcinoma: Expert review. *Gastroenterology* 2019, 156, 2149–2157. [CrossRef] [PubMed]
- 24. Tapper, E.B.; Lok, A.S. Use of Liver Imaging and Biopsy in Clinical Practice. N. Engl. J. Med. 2017, 377, 756–768. [CrossRef]

- Myers, R.P.; Pomier-Layrargues, G.; Kirsch, R.; Pollett, A.; Beaton, M.; Levstik, M.; Duarte-Rojo, A.; Wong, D.; Crotty, P.; Elkashab, M. Discordance in fibrosis staging between liver biopsy and transient elastography using the Fibroscan XL probe. *J. Hepatol.* 2012, 56, 564–570. [CrossRef]
- Carrat, F.; Fontaine, H.; Dorival, C.; Simony, M.; Diallo, A.; Hezode, C.; De Ledinghen, V.; Larrey, D.; Haour, G.; Bronowicki, J.-P.; et al. Clinical outcomes in patients with chronic hepatitis C after direct-acting antiviral treatment: A prospective cohort study. *Lancet* 2019, 393, 1453–1464. [CrossRef]
- Pons, M.; Rodriguez-Tajes, S.; Esteban, J.I.; Marino, Z.; Vargas, V.; Lens, S.; Buti, M.; Augustin, S.; Forns, X.; Mínguez, B.; et al. Non-invasive prediction of liver-related events in patients with HCV-associated compensated advanced chronic liver disease after oral antivirals. *J. Hepatol.* 2020, 72, 472–480. [CrossRef]
- Kozbial, K.; Moser, S.; Al-Zoairy, R.; Schwarzer, R.; Datz, C.; Stauber, R.; Laferl, H.; Strasser, M.; Beinhardt, S.; Stättermayer, A.F.; et al. Follow-up of sustained virological responders with hepatitis C and advanced liver disease after interferon/ribavirin-free treatment. *Liver Int.* 2018, *38*, 1028–1035. [CrossRef]
- Romano, A.; Angeli, P.; Piovesan, S.; Noventa, F.; Anastassopoulos, G.; Chemello, L.; Cavalletto, L.; Gambato, M.; Russo, F.P.; Burra, P.; et al. Newly diagnosed Hepatocellular Carcinoma in patients with advanced hepatitis C treated with DAAs: A prospective population study. *J. Hepatol.* 2018, 69, 345–352. [CrossRef]
- Calvaruso, V.; Cabibbo, G.; Cacciola, I.; Petta, S.; Madonia, S.; Bellia, A.; Tinè, F.; Distefano, M.; Licata, A.; Giannitrapani, L.; et al. Incidence of hepatocellular carcinmoma in patients with HCV-associated cirrhosis treated with direct-acting antiviral agents. *Gastroenterology* 2018, 155, 411–421. [CrossRef]
- Marino, Z.; Darnell, A.; Lens, S.; Sapena, V.; Diaz, A.; Belmonte, E.; Perellò, C.; Calleja, J.L.; Varela, M.; Rodriguez, M.; et al. Time association between hepatitis C therapy and Hepatocellular Carcinoma emergence in cirrhosis: Relevance of non-characterized nodules. *J. Hepatol.* 2019, *70*, 874–884. [CrossRef] [PubMed]
- Ogawa, E.; Furusyo, N.; Nomura, H.; Dohmen, K.; Higashi, N.; Takahashi, K.; Kawano, A.; Azuma, K.; Satoh, T.; Nakamuta, M.; et al. Short-term risk of Hepatocellular Carcinoma after hepatitis C virus eradication following direct-acting anti-viral treatment. *Aliment. Pharmacol. Ther.* 2018, 47, 104–113. [CrossRef] [PubMed]
- Nahon, P.; Layese, R.; Bourcier, V.; Cagnot, C.; Marcellin, P.; Guyader, D.; Pol, S.; Larrey, D.; De Lédinghen, V.; Ouzan, D.; et al. Incidence of Hepatocellular Carcinoma after direct antiviral therapy for HCV in patients with cirrhosis included in surveillance programs. *Gastroenterology* 2018, 155, 1436–1450. [CrossRef] [PubMed]
- 34. Kanwal, F.; Kramer, J.; Asch, S.M.; Chayanupatkul, M.; Cao, Y.; El-Serag, H.B. Risk of hepatocellular cancer in HCV patients treated with direct-acting antiviral agents. *Gastroenterology* **2017**, *153*, 996–1005. [CrossRef]
- 35. Mecci, A.J.; Kemos, P.; Leen, C.; Lawson, A.; Richardson, P.; Khakoo, S.I.; Agarwal, K.; Mutimer, D.; Rosenberg, W.M.; Foster, G.R.; et al. The association between Hepatocellular Carcinoma and direct-acting anti-viral treatment in patients with decompensated cirrhosis. *Aliment. Pharmacol. Ther.* **2019**, *50*, 204–214. [CrossRef]
- Singer, A.W.; Reddy, K.R.; Telep, L.E.; Osinusi, A.O.; Brainard, D.M.; Buti, M.; Chokkalingam, A.P. Direct-acting antiviral treatment for hepatitis C virus infection and risk of incident liver cancer: A retrospective cohort study. *Aliment. Pharmacol. Ther.* 2018, 47, 1278–1287. [CrossRef]
- Schlesinger, S.; Aleksandrova, K.; Pischon, T.; Fedirko, V.; Jenab, M.; Trepo, E.; Boffetta, P.; Dahm, C.; Overvad, K.; Tjonneland, A.; et al. Abdominal obesity, weight gain during adulthood and risk of liver and biliary tract cancer in an European cohort. *Int. J. Cancer* 2013, 132, 645–657. [CrossRef]
- Ciancio, A.; Ribaldone, D.G.; Dotta, A.; Giordanino, C.; Sacco, M.; Fagoonee, S.; Pellicano, R.; Saracco, G.M. Long-term follow-up of diabetic and non diabetic patients with chronic hepatitis c successfully treated with direct acting antiviral agents. *Liver Int.* 2021, 41, 276–287. [CrossRef]
- Cucchetti, A.; Cescon, M.; Erroi, V.; Pinna, A.D. Cost-effectiveness of liver cancer screening. *Best Pract. Res. Clin. Gastroenterol.* 2013, 27, 961–972. [CrossRef]
- Farhang Zangneh, H.; Wong, W.W.L.; Sander, B.; Bell, C.M.; Mumtaz, K.; Kowgier, M.; van der Meer, A.J.; Cleary, S.P.; Janssen, H.L.A.; Chan, K.K.W.; et al. Cost-effectiveness of Hepatocellular Carcinoma surveillance after a sustained virologic response to therapy in patients with hepatitis C virus infection and advanced fibrosis. *Clin. Gastroenterol. Hepatol.* 2019, 17, 1840–1849. [CrossRef] [PubMed]
- Goossens, N.; Singal, A.G.; King, L.Y.; Andersson, K.L.; Fuchs, B.C.; Besa, C.; Taouli, B.; Chung, R.T.; Hoshida, Y. Cost-effectiveness of risk score-stratified Hepatocellular Carcinoma screening in patients with cirrhosis. *Clin. Transl. Gastroenterol.* 2017, *8*, e101. [CrossRef]
- Mueller, P.P.; Chen, Q.; Ayer, T.; Nemutlu, G.S.; Hajjar, A.; Bethea, E.D.; Peters, M.L.B.; Lee, B.P.; Janjua, N.Z.; Kanwal, F.; et al. Duration and cost-effectiveness of Hepatocellular Carcinoma surveillance in hepatitis C patients after viral eradication. *J. Hepatol.* 2022, 77, 55–62. [CrossRef] [PubMed]
- 43. Caviglia, G.P.; Troshina, G.; Santaniello, U.; Rosati, G.; Bombaci, F.; Birolo, G.; Nicolosi, A.; Saracco, G.M.; Ciancio, A. Long-term Hepatocellular Carcinoma development and predictive ability of non-invasive scoring systems in patients with HCV-related cirrhosis treated with direct-acting antivirals. *Cancers* 2022, 14, 828. [CrossRef] [PubMed]
- Casadei Gardini, A.; Foschi, F.G.; Conti, F.; Petracci, E.; Vukotic, R.; Marisi, G.; Buonfiglioli, F.; Vitale, G.; Ravaioli, F.; Gitto, S.; et al. Immune inflammation indicators and ALBI score to predict liver cancer in HCV-patients treated with direct-acting antivirals. *Dig. Liver Dis.* 2019, *51*, 681–688. [CrossRef] [PubMed]

- 45. Abe, K.; Wakabayashi, H.; Nakayama, H.; Suzuki, T.; Kuroda, M.; Yoshida, N.; Tojo, J.; Kogure, A.; Rai, T.; Saito, H.; et al. Factors associated with Hepatocellular Carcinoma occurrence after HCV eradication in patients without cirrhosis or with compensated cirrhosis. *PLoS ONE* **2020**, *15*, e0243473. [CrossRef]
- 46. Tanaka, Y.; Ogawa, E.; Huang, C.F.; Toyoda, H.; Jun, D.W.; Tseng, C.H.; Hsu, Y.-C.; Enomoto, M.; Takahashi, H.; Furusyo, N.; et al. HCC risk post-SVR with DAAs in East Asians: Findings from the REAL-C cohort. *Hepatol. Int.* **2020**, *14*, 1023–1033. [CrossRef]
- 47. Van der Meer, A.J.; Feld, J.J.; Hofer, H.; Almasio, P.L.; Calvaruso, V.; Fernandez-Rodriguez, C.M.; Aleman, S.; Ganne-Carrié, N.; D'Ambrosio, R.; Pol, S.; et al. Risk of cirrhosis-related complications in patients with advanced fibrosis following hepatitis C virus eradication. J. Hepatol. 2017, 66, 485–493. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.